Author's response to reviews

Title: The roles of tumor necrosis factor-alpha in colon tight junction protein expression and intestinal mucosa structure in a mouse model of acute liver failure

Authors:

Hongli Song (hlsong26@yahoo.com.cn)
Sa Lv (lysa@sina.com)
Pei Liu (syliupei2003@yahoo.com.cn)

Version: 7 Date: 18 September 2009

Author's response to reviews: see over
Dear editor:

We thank you for the opportunity to publish our manuscript (MS: 1297437313267928) which has now been revised in response to the comments provided by the BioMed Central Editorial Production Team. We have been able to attend to the majority of the points raised and we hope that the revised version of the manuscript is acceptable for publication in the BioMed Central. The following are point-to-point answer to the BioMed Central Editorial Production Team’ comments.

If you still have some question on revision, please contact us without any hesitation.

Thanks a lot.

Sincerely yours

Hongli Song, Dr.
Department of Infectious Diseases,
The First Affiliated Hospital, China Medical University
Present address: Department of Organ Transplantation,
Tianjin First Central Hospital, Tianjin
Tel: +8613820966276
Email: hlsong26@yahoo.com.cn

Professor Pei Liu
Department of Infectious Disease
First Affiliated Hospital of China Med. University.
**Major revisions** (MS: 1297437313267928)

**Comment 1:** Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

**Response:** We have revised the name exactly in the submission system and written them identically.

**Comment 2:** Tables as additional files - We notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

**Response:** We have revised the tables in the manuscript following the references. Thank you very much.

**Minor revisions** (MS: 1297437313267928)

**Comment 1.** Please connect the authors to their affiliations with superscript numbers and remove the duplication of the author names

**Response:** We have revised this word. Thank you very much.

**Comment 2.** Affiliations - For affiliations that have been up-dated since writing the paper, please include the following text "Current address:" Please avoid the use of "Present address:“, ”New address"

**Response:** We have revised this word. Thank you very much.

**Comment 3** Running head. Please remove this.

**Response:** We have remove the running head.

**Comment 4** Please place the Abbreviations directly before the Competing interests section and remove them from the Abstract
**Response:** We have placed the Abbreviations directly before the Competing interests section and remove them from the Abstract. Thank you very much.

**Comment 5** Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

**Response:** Thank you very much and I apologize for my carelessness. I have revised any typographical errors in the manuscript carefully, in the discussion section especially and this manuscript has been checked by a native speaker.